

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Link, Susanne

Working Paper Single-Sex Schooling and Student Performance: Quasi-Experimental Evidence from South Korea

ifo Working Paper, No. 146

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Link, Susanne (2012) : Single-Sex Schooling and Student Performance: Quasi-Experimental Evidence from South Korea, ifo Working Paper, No. 146, ifo Institute - Leibniz Institute for Economic Research at the University of Munich, Munich

This Version is available at: https://hdl.handle.net/10419/73700

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Working Papers

Single-Sex Schooling and Student Performance: Quasi-Experimental Evidence from South Korea

Susanne Link

Ifo Working Paper No. 146

October 2012

An electronic version of the paper may be downloaded from the Ifo website www.cesifo-group.de.

Single-Sex Schooling and Student Performance: Quasi-Experimental Evidence from South Korea*

Abstract

To obtain reliable estimates of the effects of single-sex education, I exploit the random assignment of students to single-sex and coeducational schools in South Korea. The results suggest that single-sex schooling is beneficial for girls in math, but has no effects for boys. Moreover, comparisons within and across gender reveal that girls with low supporting parental backgrounds at coeducational schools fall behind their peers which is partly explained by a rougher classroom climate at mixed schools. Several robustness checks confirm these results.

JEL Code: I20, I24, J16. Keywords: Single-sex education, student performance, random assignment, peer effects.

> Susanne Link Ifo Institute – Leibniz Institute for Economic Research at the University of Munich Poschingerstr. 5 81679 Munich, Germany Phone: +49(0)89/9224-1698 link@ifo.de

* I like to thank seminar participants at the Ifo Institute for Economic Research, in particular Ludger Woessmann, Oliver Falck, Guido Schwerdt and Nadine Fabritz, for helpful comments and discussion. I gratefully acknowledge financial support from the Deutsche Forschungsgemeinschaft through GRK 801.

1 Introduction

Underrepresentation of women in both high-paying and science-related fields is well documented. Explanations include gender-stereotype sorting, as well as differences in individual preferences, in non-cognitive behavior or in cognitive skills between men and women. Differences in the distribution of quantitative skills between boys and girls partly explain the sorting of men and women into high-paying and low-paying fields.¹ However, variation across cultures suggests that this gap is due to the social environment rather than inherent gender traits (Guiso et al., 2008).² Thus, raising girls' interest and achievement in math and sciences is a goal of policy aimed at reducing gender-based disparities. In this context, single-sex schooling has gained particular attention. For example, in the United States, single-sex classrooms are a growing phenomenon and amendments to Title IX that explicitly allow single-sex public schools and classes have set off a pedagogical dispute over whether sex-segregation improves educational achievement (Cohen, 2012; Whitmore, 2005).³

This paper exploits the random assignment procedure of students to South Korean (hereafter Korean) middle schools to investigate the effects of single-sex schooling on academic achievement in two stereotypically male subjects, namely, math and science. Given that attendance at single-sex schools is orthogonal to student characteristics such as socioeconomic background and ability, the comparison between girls (boys) at coeducational schools and girls (boys) at single-sex schools should identify a reliable effect of single-sex schooling on student achievement. By using TIMSS data from 1999, that provide extensive background information, I am able to investigate potential channels.

I find positive effects of single-sex schooling for girls at middle schools in math, but not in science. The effects in math are not only highly statistically significant and non-negligible in their magnitude, but also highly relevant since math performance is consistently linked to future earnings (Paglin and Rufolo, 1990). In contrast, I do not find any effects for boys.⁴ Several robustness checks confirm that the results are not driven by observable or unobservable differences in the types of students that attend single-sex and coeducational schools.

Comparisons within and across gender reveal that girls with non-supporting parental backgrounds at coeducational schools fall behind their peers, a finding partly explained by a rougher classroom atmosphere at mixed schools. Coleman (1961) was the first to

¹Paglin and Rufolo (1990) show that there is a much higher proportion of men than women in the top intervals of mathematical reasoning ability, which is often a qualification in high-paying fields. Interestingly, women with high mathematical reasoning abilities also show high participation rates in the sciences.

²See, e.g., Fryer and Levitt (2010) for an analysis of the gender gap in mathematics for the United States. Booth et al. (2011) emphasize the importance of social learning rather than inherent gender traits for observed gender differences in risk behavior.

³For example, Billger (2009) studies the effects of single-sex schooling in the context of the increase in single-sex classes and schooling in the United States as a response to amendments to Title IX.

⁴This finding is in line with earlier research on the effects of single-sex education. For example, Jackson (2002) finds positive effects of all-girl classes but no effects for all-boy classes.

hypothesize that the presence of the opposite sex in the classroom is distracting and leads to lower educational achievement for both boys and girls. In line with this, single-sex schools are claimed to have more serious and studious classroom climates (Lee and Bryk, 1986). This might be especially beneficial for girls given that boys are more disruptive, restless, and dominant in class. In fact, larger shares of girls in class are found to be associated with higher academic achievement which can partly be explained by a lower level of classroom disruption and violence (Hoxby, 2000; Whitmore, 2005; Lavy and Schlosser, 2011).⁵

Despite a great deal of work on the subject, empirical evidence regarding the effects of single-sex schooling on student outcomes is inconclusive (Bigler and Signorella, 2011). Several studies report positive effects, especially for girls, on academic achievement, self-esteem, and other non-cognitive outcomes (e.g. Lee and Bryk, 1986; Riordan, 1990; Jackson, 2002; Eisenkopf et al., 2011). However, other studies find no significant differences between students at coeducational and single-sex schools (Marsh, 1989). Moreover, most of the literature is based on comparison of student outcomes at coeducational schools and single-sex schools. These results are likely to be biased by self-selection of students into single-sex schools, since attendance at single-sex-schools is usually correlated with unobservable, individual characteristics that also determine student achievement.⁶

In recent years, there has been a growing literature that addresses the selection issues as to isolate the causal effect of single-sex schooling on student outcomes. Jackson (2012) exploits the fact that assignment rules in Trinidad and Tobago create exogenous variation to remove selection bias. He shows that only girls with stated preferences for single-sex schooling actually perform better. However, for most students he finds no significant effects. Eisenkopf et al. (2011) report positive effects of single-sex education for girls at a Swiss high school where girls are randomly assigned to single-sex and coeducational classrooms. In a similar manner, Behrman et al. (2012) make use of a unique feature in the Korean education system, namely, the random allocation of students to high schools in Seoul. They show that attending a single-sex school is associated with higher test scores in Korean and English and a higher probability of attending a four-year college for both girls and boys.

This paper contributes to the growing quasi-experimental literature on the effects of single-sex schooling. The random assignment of students to Korean schools presents a nice opportunity to obtain unbiased estimates. In contrast to Behrman et al. (2012), I focus on middle schools which are compulsory and therefore represent the full population of eighth grade students. Moreover, investigating the effects of single-sex schooling on math and sciences is especially interesting given the discussion about the influence of gender stereotypes on student achievement and choice (Thompson, 2003; Joshi et al., 2010;

⁵However, Whitmore (2005) reports that the positive effects for boys are only found for low grade level. ⁶See, e.g., Lee and Bryk (1986) for an overview of reasons to choose single-sex schools over coeducational schools.

Favara, 2011). Further, employing data provided by TIMSS and PISA is advantageous. Using externally assessed test scores in contrast to teacher assessed grades reduces the risk that effects are driven by teacher discrimination against girls or boys or by grading relative to peer performance. Moreover, the rich background questionnaire used in this study allows me to investigate a broad set of potential channels, which is an additional contribution to the literature.

My in-depth analysis shows that the positive effects for girls are neither explained by differences in school and teacher characteristics at coeducational and single-sex schools nor by gender-tailored teaching practices or more positive attitudes toward math at single-sex schools. However, some of the effect can be attributed to a rougher classroom atmosphere at mixed schools. The fact that I cannot fully eliminate the positive effect of single-sex schooling suggests that girls with non-supporting family backgrounds somehow benefit from the absence of boys in class. The more general implication may be that in any school system, girls with a non-supporting background may be particularly influenced by less favorable peer characteristics.⁷

The remainder of the paper is organized as follows: Section 2 describes the random assignment process to Korean schools. Section 3 presents the data. Section 4 explains the empirical approach. Section 5 provides the main results, along with an analysis of potential channels and mechanisms. Section 6 details the results of several robustness checks. Section 7 summarizes and concludes.

2 The Random Assignment Process to Korean Schools

As a response to a fierce competition among students in the admission process to middle and high schools, an "Equalization Policy" (EP) was introduced in 1969 with the aim of creating equal education opportunities at middle schools and reducing the influence of social background on student educational achievement.⁸ Under this policy, the competitive entrance examinations were replaced by a random allocation, via a lottery system, of students within each school district. In other words, all schools, regardless of whether they were public or private, could no longer select students themselves but instead were required to take all students assigned to them by the Ministry of Education via a district-wide lottery. Moreover, the policy required equalization of school resources and teachers in an effort to ensure that there were no differences in resources and instruction quality across schools (Kim and Lee, 2003). Curriculum and teacher qualifications became uniform and centrally regulated. The government even provided subsidies to financially weak private schools so that their teacher salaries would equal those of public schools. Furthermore, all

⁷The heterogeneity of peer effects across gender is also documented by Lavy et al. (2012) that show that only girls significantly benefit from the presence of academically strong peers, the presence of very academically bright peers.

⁸See, e.g., section 7 in the Appendix for more information and Kim and Lee (2003) for an overview of the Korean education system in general and education reforms in particular.

private schools were required to charge the same tuition and teach the central curriculum.⁹

The policy was first implemented in Seoul in 1969 and expanded to major cities and then throughout the entire country within the next two years. Differences in teacher quality and school resources between schools were quickly reduced and improvements in the physical and psychological development of children reported. However, now that the problem was solved at the middle school level, an even fiercer competition for prestigious high schools began. As a response, the government introduced the high school Equalization Policy in 1974 for general high schools.¹⁰ Under the policy, entrance examinations for general high schools were abolished. After passing a screening process, applicants for general high schools were assigned by lottery to a school within their residential district. Again, the policy was first adopted in Seoul and Pusan, the two largest Korean cities. By 1980, the Equalization Policy had been expanded to cover most major Korean cities.

The original structure of the Equalization Policy has been maintained for the past 30 years, leaving its main guidelines unchanged. Even today, all middle school students are assigned by lottery to a school within their residential district (Lee, 2004; Kim and Lee, 2003). However, the high school Equalization Policy became the subject of discussion and critique during the 1990s. As a result, its implementation was slowed. Metropolitan cities continued to be required to follow the policy and assign their students to general high schools, but it was optional for smaller cities and rural school districts. In 2001, the high school EP covered all seven metropolitan areas and 11 provincial cities. This accounts for 51 percent of the country's 1,969 high schools and 65 percent of their 1.91 million total students (Kim et al., 2008). More recently, some school districts modified the Equalization Policy such that students are allowed to state their preferences and high schools may choose a fraction of their students.

3 Data

Secondary schooling in Korea is organized into lower and upper secondary education. After graduating from middle school, which ranges from Grade 7 to 9, students usually proceed to upper secondary education (ISCED 3) and attend either general or vocational high schools.¹¹ Graduates of vocational high schools are qualified for direct entry into the labor market. In contrast, general high schools are more academically oriented and

⁹Except for certain rights over personnel decisions and school facilities, there are no differences between private and public schools in Korea. Even essential features of private schools, such as selection of students, tuition fees, teacher salaries, and curriculum, are regulated. Because of their short history, private schools in Korea were less prestigious and most of them welcomed the financial support and assignment of better students.

¹⁰High schools in Korea are divided into general, more academically oriented high schools, and vocational high schools that qualify for direct entry into the labor market. Vocational high schools have always been excluded from the Equalization Policy. Applicants for vocational high schools are allowed to state their preferences and are then selected by schools based on entrance examinations or middle school results.

¹¹There are also a few specialized schools which are not considered in this study.

qualify their graduates for tertiary education. Interestingly, Korea has a long-standing tradition of single-sex schooling, with about half the students attending single-sex schools in 2000. It was not until the 1980s that coeducational schools emerged and conversion of some single-sex schools followed. Since the Equalization Policy for middle schools was not modified subsequent to this development, students are randomly assigned within the boundaries of their school districts regardless of whether schools are single-sex schools or coeducational schools.

To give an overview of the population of students, I report summary statistics on the eighth grade students at middle school and the 15-year-old students at high school by using data provided by the Third International Mathematics and Science Study (TIMSS) and the Programme for International Student Assessment (PISA). TIMSS 1999 tested countries in the southern hemisphere between September and November 1998, while PISA 2000 tested the majority of students in April 2000. Since the Korean school year starts in March and ends in mid-February, I observe the same cohort of students at the end of Grade 8 at middle school and at the beginning of Grade 10 at high school by using the TIMSS 1999 and the PISA 2000 data.

Besides educational achievement indicators, both datasets provide extensive background information at the student level as well as information on school and teacher characteristics. To indicate the single-sex status of a school, I rely on information as to the number of girls and boys enrolled at a school.¹² I drop observations from villages or rural areas because those areas are likely to have only a limited number of schools to which students could be assigned. In other words, by restricting the sample to large towns and cities, I focus on areas where the average school district has several coeducational as well as several all-girl and all-boy schools. For instance, in a typical school district (Kangnam) within the capital of Seoul, there are 10 coeducational schools, seven all-boy schools and seven all-girl schools to which students can be assigned (Seoul Metropolitan Office of Education 2007). The resulting dataset totals 4,775 individual observations at middle schools and 4,390 individual observations at high schools.

Given that I am interested in the effect of single-sex schooling on student achievement, there are a number of reasons to focus on middle schools only. Most important, all students are assigned randomly to middle schools according to the Equalization Policy. In contrast, vocational schools are not targeted by the EP and general high schools are subject to a number of exceptions. Moreover, the TIMSS data comprise the full population of students attending the eightht grade. In contrast, students select themselves into either vocational or general high schools after graduating from middle school. Figure 1 shows that the proportions of girls and boys at middle schools are quite equal — presumably because middle school is compulsory — whereas the share of boys exceeds the share of girls at

¹²The number of boys and girls enrolled at school is only collected in TIMSS 1995 and 1999. Since about 30 percent of students at coeducational schools are taught in single-sex classes in TIMSS 1999, I do not infer single-sex school status by the share of girls in a class in more recent waves of TIMSS and use TIMSS 1999, only.

general high schools. This is due to the fact that a relatively large share of girls attends vocational high schools, which are less academically oriented, or even drops out of school after graduating from middle school.

Furthermore, students at middle schools are nearly equally divided between single-sex and coeducational schools. About one half of all girls and all boys attend all-girl or all-boy schools, respectively. The figures for high school and vocational school students, however, reveal a disproportionate distribution of students between single-sex and coeducational schools. This indicates that the number of all-girl, all-boy and coeducational high schools is not evenly distributed. In addition, I observe students at a later point in time in their middle school career. Assuming that a single-sex school effect needs some time to unfold, I am more likely to find any effects for students at the end of their second year at middle school compared to students at the beginning of their first year at high school.

Tables 1 and 2 report student characteristics separately by gender for single-sex and coeducational middle schools. This comparison is intended to provide a first indication of the extent of randomness in the allocation process to school types. If students are randomly assigned to schools, student characteristics should not differ across single-sex and coeducational schools. The figures by gender and school affiliation are very similar and differences are generally not statistically significant for conventional student background characteristics. Unfortunately, I cannot observe residential school districts, and the very few significant differences in student characteristics are possibly driven by differences in the location of schools. However, several robustness checks suggest that the results are not driven by differences in the population of students between single-sex and coeducational schools.

4 Empirical Strategy

In the literature, the effects of attending a single-sex school are mostly derived by comparing students at coeducational and single-sex schools while controlling for a rich set of background variables. However, these estimates are unbiased only if the variable of interest, attendance at a single-sex school, is not correlated with unobservable characteristics captured by the error term. To satisfy this assumption, recently a number of studies make use of quasi-experimental settings (Eisenkopf et al., 2011; Behrman et al., 2012; Jackson, 2012).

To obtain the effect of single-sex schooling on student performance, I estimate the following model

$$TS_{ic} = \alpha + \beta SS_c + \gamma' X_{ic} + \varepsilon_i + \eta_c.$$
⁽¹⁾

 TS_{ic} is student *i*'s performance at school *c* in either math or science, while SS_c indicates if student *i* is attending a single-sex school (1, if single-sex). The dependent variable is

normalized with a mean of 0 and a standard deviation of 1. X_{ic} denotes a large set of control variables at the individual, school, and teacher level, ε_i represents an idiosyncratic error term, and η_c the error component that varies at the school level. In all regressions, I cluster standard errors at the school level to account for the fact that students at the same school share similar background and identical school and teacher characteristics.

As mentioned above, the interpretation of β relies on the underlying assumption that attendance at single-sex schools is orthogonal to unobserved individual characteristics. Since all middle schools are covered by the Equalization Policy, all middle school students are randomly assigned to a school within their residential district and neither observables nor unobservable characteristics should bias my estimates. I run all regressions separately for girls and boys, implicitly comparing girls (boys) at single-sex schools with girls (boys) at coeducational schools. By gradually adding control variables, I check whether differences in student characteristics, family background, and school and teacher characteristics alter the estimates. Since the random assignment process should be reflected in very similar background characteristics, adding information on student's socioeconomic background should not reverse or fundamentally alter my estimates. In contrast, differences in school resources and teacher characteristics can be seen as potential channels through which single-sex schooling affects student achievement. Although the Equalization Policy aimed at the equalization of schools and the ultimate reduction of differences in school quality, there are small differences between single-sex and coeducational schools in standard school and teacher characteristics (see Table A.1 in the Appendix).

The baseline model is then extended. To investigate the underlying mechanisms, I account for differences in the disciplinary classroom climate, teaching practices, and student attitudes, all of which are often argued to be influential determinants in the public debate. Further, I check whether effects of single-sex education are heterogenous across student groups and divide my sample implicitly into students from *relatively supporting* and *relatively non-supporting* families. Finally, I compare students across gender to see how girls at single-sex and coeducational schools actually perform relative to their male peers.

5 The Effect of Single-Sex Schooling on Student Performance

5.1 Baseline Results

Table 3 reports OLS estimates on the effect of single-sex schooling for girls and boys at middle schools. I start with a univariate model with *attending a single-sex school* as the explanatory variable. The model is then — step by step — extended by controls on student

and background characteristics, school characteristics, and teacher characteristics.¹³ The variable of interest, attending a single-sex school, is positive and significant in math for girls at middle schools throughout all specifications. Neither adding individual control variables (Column 2) nor school (Column 3) and teacher variables (Column 4) alters magnitude and significance. In other words, the coefficient is robust to the inclusion of control variables, which in turn suggests that attendance at single-sex schools is not correlated with unobservables that affect both control variables and outcome. Overall, these results suggest that girls at single-sex schools outperform girls at coeducational schools by about 13.5 percent (Column 4). For science, I find a positive coefficient that is not significant at conventional levels throughout the specifications.

The lower part of Table 3 reports the results for boys. I find insignificant coefficients, which are mainly close to zero for all specifications and both subjects. This indicates that there are no beneficial effects of single-sex schooling for boys at middle schools.¹⁴

Since students start middle school in Grade 7 and I observe them at the end of Grade 8, the effects I find for girls in math are likely to be cumulative.¹⁵ TIMSS tests elements of primary and secondary school curriculum (Hanushek and Woessmann, 2011).¹⁶ Compared to science, which is not taught as a single subject at most schools in my sample, math achievement might be a better indicator of teacher instruction in class. Thus, if more studious classrooms allow teachers at all-girl schools to cover the curriculum more extensively, this might be reflected in the large, significant coefficients for math achievement. Moreover, math is a traditionally male subject. The positive effects at single-sex school might also be driven by less gender-stereotyped attitudes.

In the upcoming analysis, I investigate several issues raised in the discussion of my findings. First, I focus on channels that might explain the positive effects for girls at single-sex schools. Second, I investigate whether the effects are limited to a specific group of students and compare girls' achievement relative to that of boys.

5.2 Channels of the Effects of Single-Sex Schooling

Given the positive and large effects for girls in math, it is important to understand the mechanisms that drive these effects. In a larger sense, studies on single-sex schooling contribute to the literature on peer effects which deals with the effects of all sorts of peer characteristics, including gender, on academic achievement (Hoxby, 2000; Lavy and Schlosser, 2011). Thus, it is important to separate gender compositional effects from other peer effects, such as advantageous family backgrounds and environments. Even though students in Korea are randomly assigned to schools within their residential districts, it

 $^{^{13}}$ See Table A.2 for the complete model.

¹⁴Throughout the paper, I report estimation results using the first plausible value reported in the data. However, the results are robust to using other plausible values or the mean of all plausible values reported. ¹⁵See Heckman (2006) for evidence on life-cycle skill formation.

¹⁶For a comparison on international student assessment tests, see Neidorf et al. (2006).

is possible, but rather unlikely, that I observe only all-girl schools in better-off areas, whereas I observe coeducational schools in disadvantaged areas. If girls with advantageous family backgrounds are grouped within all-girl schools, the effects I find are not due to the absence of boys, but could be attributed to a better student composition at single-sex schools. For example, Jimenez and Lockheed (1989) attribute positive effects of single-sex schooling for girls in Thailand to favorable peer characteristics, rather than gender.

To make sure that it is the absence of boys in contrast to advantageous family background characteristics of female peers that drive the positive effects for girls, I account for the quality of a student's peers in the regression. This approach is especially comprehensive, since TIMSS tests complete classes in math. Thus, in Table 4, I control for the share of peers with a low socioeconomic background as measured by the books at home, the share of peers with high family resources as a proxy for wealth, the average family size of an individual's peers, the share of peers with at least one parent holding a university degree, the share of peers with mothers who hold an university degree, and the average amount of time spent studying by an individual's peers as a proxy for peer pressure. None of these controls change the estimate of single-sex schooling for girls or boys, suggesting that the effects are not due to selection, but to factors such as classroom interaction and climate that are different in single-sex schools.

Since the equalization of resources across schools is one part of the EP, the results should also not be driven by conventional school characteristics. Table 3 shows that controlling for conventional school characteristics does not change the estimates. However, single-sex and coeducational schools might differ in the atmosphere and organization within schools. Even though some of these dimensions are unobservable, I am able to compare coeducational and single-sex schools in three influential areas — teaching practices, student attitudes toward math, including self-perceived competence, and disciplinary climate — to check whether the positive effect of single-sex schools for girls can be explained by differences between single-sex and coeducational schools.

The most obvious reason why single-sex education might be especially beneficial for girls involves the relatively more restless and disruptive behavior of boys. A growing body of literature documents that a larger share of boys in a class is associated with lower academic achievement (Hoxby, 2000; Lavy and Schlosser, 2011). TIMSS reports students', teachers', and principals' perceptions on several aspects of the disciplinary climate of classrooms and schools. Table A.3 in the Appendix reveals that, according to teacher and principal reports, there are indeed differences in the disciplinary climate at coeducational, all-girl, and all-boy schools. Teachers are asked to what extent teaching is hindered by (1) disruptive students, (2) uninterested students, (3) a wide range of backgrounds, and (4) a wide range of academic abilities. Twenty-five to 28 percent of students at coeducational schools attend classrooms where "disruptive" and "uninterested" students are reported to be "a serious problem". These fractions are somewhat smaller for all-boy schools and about half the size at all-girl schools. Further, teachers at coeducational schools perceive "a wide range of backgrounds" and "a wide range of academic abilities" as a problem more often compared to teachers at single-sex schools. Moreover, at more than 60 percent of all-girl schools, the "injury of students" is "not a problem at all", but there are large fractions of coeducational and all-boy schools that report the it as "quite problematic". This indicates that the disciplinary climate, as reported by both teachers and principals, is rougher at coeducational schools, which might be especially detrimental to girls' achievement.

Another argument made in favor of single-sex education is that such schools offer the opportunity to tailor schooling to each sex's unique needs. Differences in the way students are taught, therefore, might account for the positive effects found for girls. On the one hand, supporters of single-sex education claim that brain differences between boys and girls require different teaching styles.¹⁷ On the other hand, more studious classroom climates at all-girl schools might motivate teachers to give more homework or work more often in groups.¹⁸ Table A.4 in the Appendix shows that students at both all-girl and all-boy schools more often report "copying notes from the board" compared to students at coeducational schools. Students from all-girl schools report that teachers "give homework more frequently" and that they "work more often in groups". However, the reported differences are quite small.

Student attitudes toward math present another possible channel and the one most closely related to the literature on gender stereotypes. The construction of gender identities at schools seems especially important with regard to the persisting gender test score gap in math (see, e.g., Guiso et al., 2008; Fryer and Levitt, 2010) and the low representation of women in math- or science- related fields.¹⁹ The presence of the opposite sex at mixed schools may either deforce or reinforce gender-stereotyped attitudes and thereby influence the likelihood that boys (girls) engage in stereotypically female (male) subjects or fields.²⁰ Single-sex education may reduce gender stereotype attitudes, it may lead to gender-atypical educational choices, and it might increase girls' interest in math, which is likely to improve learning and achievement (e.g. Thompson, 2003; Joshi et al., 2010; Favara, 2011). Moreover, the presence of boys in the classroom could be especially intimidating for girls in a stereotypically male subject such as math. Given a predominant opinion that boys outperform girls in math, a girl at a coeducational school is more likely to assess herself poorly relative to her peers, which include girls **and** boys, compared to a girl at a single-sex school.²¹ Table A.5 reports descriptive statistics on several indicators of student

 $^{^{17}\}mathrm{Neuroscientists}$ have found only few differences and none of them have been linked to teaching practices.

¹⁸The fact that achievement gains can be driven by differences in teaching styles has been documented by Jürges and Schneider (2010) who attribute positive effects of central exit exams to the fact that students were required to work harder.

¹⁹The role of gender identities is based on Akerlof and Kranton (2000)

 $^{^{20}}$ For example, Favara (2011) confirms that subject choices of girls at single-sex schools are more similar to those of their male schoolmates. In contrast, Halpern et al. (2011) show that sex segregation increases gender stereotyping.

 $^{^{21}\}mathrm{Beyer}$ and Bowden (1997) show that females' self-perceptions of performance were inaccurately low in male tasks.

attitudes toward math, their self-perceived competence in math, and their educational aspirations. Girls at coeducational schools seem to have a less positive attitude toward math compared to girls at single-sex schools; apart from that, however, there are only very small differences regarding their confidence, educational aspirations, and preferences for math.

Classroom climate, teaching practices, and student attitudes all have the potential to interact with student learning and — as discussed above — there are plausible reasons why those areas might differ between single-sex schools and coeducational schools. However, except for the disciplinary climate, the descriptive statistics report only very small — if any — differences between coeducational and single-sex schools. Nevertheless, I control for disciplinary climate, teaching practices, and student attitudes in the regressions. If the coefficient of interest, single-sex schooling, is capturing some of these differences, the coefficient should decrease in size and significance.

Table 5 shows the relationship between measures of teaching practices and student achievement. The frequency of "having tests" and "giving homework" seems to be positively associated with student learning, however, most of the other measures are insignificant. Most importantly, the coefficient on the variable *single-sex schooling* does not change in magnitude or significance for either girls or boys. This suggests that the effects of single-sex schooling are not driven by differences in teaching practices. Table 6 reports the association between several measures of student attitudes toward math and achievement. As expected, all the measures are positively and significantly associated with better math results for both, boys and girls. However, although student attitudes have strong explanatory power for student achievements, the positive effects of single-sex schooling remain significant and are not affected by differences in the attitude toward math.

The association between several measures of the disciplinary climate and student achievement is also set out in Table 7. If students "behave orderly and as told" in class, they show higher achievement. However, the coefficient of interest is not influenced by the measures reported by students. Interestingly, I find no significant association between the extent of "disruptive students" and male and female achievement. In line with this, the effect of single-sex schooling is unchanged in these specifications. Also, the coefficient of interest is not affected by the inclusion of the extent of "intimidation of students" at school. This is not surprising, since principal reports are very similar at coeducational and single-sex schools.

In contrast, Table 7 reveals a negative and strong association between teachers who report "uninterested students as a great problem" and student achievement for girls. The coefficient of single-sex schooling drops by one quarter and loses significance in the girls' regression. Similarly, the "injury of students" presents a larger problem at coeducational schools and reduces the estimate of single-sex schooling considerably in the girls' regression. Although some teachers report "differences in students' backgrounds as a problem", this is not reflected in lower achievement by students. However, teachers at coeducational schools more often report that "differences in the math abilities" of their students limit their teaching. This is also reflected in lower student achievement and reduces the estimate of single-sex schooling. Since students are randomly assigned to single-sex and coeducational schools, the variation in math ability at each type of school should initially be quite similar. One explanation for that observation, therefore, might be that achievement of students in general or of boys and girls in particular at coeducational schools has diverged over time. Alternatively, it might be that teachers of coeducational classes just perceive abilities as more diverse, possibly due to a predetermined opinion that boys outperform girls in math.

Overall, these results suggest that differences in teaching practices and student attitudes cannot explain the achievement gains for girls at single-sex schools. However, in reality, Table 7 suggests not that girls educated in a single-sex school do better, but that girls at coeducational schools do worse due to a rougher classroom atmosphere.

5.3 Heterogenous Effects of Single-Sex Schooling

In a next step, I investigate whether effects vary with student family background. Paying attention to students with relatively less supportive backgrounds is important for several reasons. First, it has been argued that either type of schooling might be more beneficial or harmful to some students than to others. For example, Riordan (1990) shows that the greatest gains in single-sex schooling are those experienced by Hispanic and African-American males and females at schools with large minority populations. One reason for this might be that students with low socioeconomic background typically receive less support at home in studying and, since their education depends more strongly on instruction received at school, respond more strongly to it. Another reason might be that students belonging to minorities — either ethnic or socioeconomic — are easily intimidated and need a great deal of attention or support. Paying attention to students from a low socioeconomic background is also politically relevant since those students are at a higher risk of dropping out of school or performing very poorly, which might come at a high cost for the society as a whole (OECD, 2009; Woessmann and Piopiunik, 2009).

The number of books at home is a strong predictor of academic achievement by both girls and boys, as Table A.2 in the Appendix reveals, and has often been used in the literature as a measure for socioeconomic background (see, e.g., Woessmann, 2003, 2008; Schütz et al., 2008). Thus, I divide the sample into two groups and classify students with less than 100 books at home as students with *relatively low socioeconomic background* and those with more than 100 books at home as students with *relatively low socioeconomic background* and those with more than 100 books at home as students with *relatively low socioeconomic background* and those with more than 100 books at home as students with *relatively low socioeconomic background*. I further generate a variable that takes the value 1 if students have relatively low educated parents since parental education is a strong indicator of parents' interest in their children's educational aspirations and development. Similarly, a variable indicating wether a student reports that his or her mother is not interested in his or her math achievement is generated. I then interact those measures with single-sex schooling and

include them in the regression.

As expected, Table 8 shows that a low socioeconomic background, low educated parents, and uninterested mothers are strongly and negatively associated with girls' and boys' math achievement. Interestingly, the interaction of all three measures of a supportive background are positive and significant for girls. Moreover, positive effects of single-sex schooling only occur for girls with low socioeconomic or low educated family backgrounds, and the effects are even larger for girls who report that their mothers are not interested in their math achievement. Consistent with the previous results for boys, the effects of single-sex schooling remain insignificant and around zero and the coefficients on the interactions are not significant.

So far, the analysis compared girls (boys) at coeducational schools with girls (boys) at single-sex schools. The results suggest that girls from low parental support backgrounds at single-sex schools outperform girls from low parental support backgrounds at coeducational school, whereas there are no significant differences for boys. However, given the existence of a gender test score gap in math, it is also interesting how girls at single-sex and coeducational schools perform relative to boys. Table 9 shows a pooled regression divided by socioeconomic background. The coefficient on the female dummy can be interpreted as the gender test score gap in math. Without controlling for single-sex schools, there is no significant difference in math achievement between boys and girls in the full sample (Column 1). However, as soon as the regression controls for all-boy (coefficient on single-sex) and all-girl schools (coefficient on the interaction of single-sex and female), the coefficient on the female dummy turns negative and significant, revealing the famous female test score gap in math. In other words, Columns 1 and 2 show that there are no significant differences in math achievement between boys at either coeducational schools or single-sex schools and girls at single-sex schools. However, girls at coeducational schools underperform boys at both school types and girls at single-sex schools. Table 9 also reports the results for students with low and high socioeconomic background as measured by books at home (Columns 4 to 9). Interestingly, there is no test score gap in math between boys and girls from relatively high socioeconomic background (Column 7), not even after controlling for single-sex schools (Column 8). In contrast, the test score gap between boys and girls with low socioeconomic background at coeducational schools is especially large (Column 5). Interestingly, the test score gap for girls in math vanishes as soon as the regression additionally controls for "the extent of injuries", — which can be viewed as a proxy for disciplinary climate. Altogether, this in-depth analysis suggests that girls from less supportive backgrounds fall behind at coeducational schools and that the atmosphere in coeducational classrooms plays an important role in this result.

6 Robustness

My results suggest that single-sex schooling seems to be beneficial in regard to math achievement for girls from low parental support backgrounds in math, but does not have any effects for boys. The causal interpretation in an ordinary least squares approach is based on the assumption that attendance at single-sex schools is orthogonal to student characteristics such as socioeconomic background and ability, and since students in Korea are randomly assigned to schools, this is very likely the case. Tables 1 and 2 lend support to this assumption by reporting very small and mostly non-significant differences in a very rich set of observable student characteristics. Moreover, the robustness of the estimates to the inclusion of this extensive set of control variables further corroborates the assumption (see Table 3).

The few significant differences in student characteristics may very well be driven by differences in the location of schools but, unfortunately, I cannot observe residential school districts. Table 4 shows that controlling for peer quality as a proxy for neighborhood characteristics does not change the estimates. Furthermore, controlling for the background variables reported in Table 1 at the class-level leaves the estimate unchanged. Nevertheless, I perform propensity score analysis and compare students at single-sex and coeducational schools who have similar estimated propensities to attend single-sex schools based on observable characteristics. I perform two common matching techniques — namely kernel and nearest-neighbor — since, to date, no single method has been found to be superior in the matching literature.²² However, both the OLS and propensity score estimates are biased and inconsistent if there are unobservable characteristics that directly affect student achievement and are also correlated with single-sex school attendance. Again, since students are randomly assigned to schools, differences in unobservable characteristics between students at single-sex and coeducational schools only emerge out of differences in school locations. In other words, I do not worry that students attending single-sex schools are in general more motivated or the like, but I cannot exclude the possibility that school districts with, for example, highly motivated citizens have more single-sex schools. Students in those highly-motivated school districts would then have a higher probability of attending single-sex schools. Given that this argument also applies to boys at single-sex schools, I am confident that differences in unobservable characteristics are not driving my results. Nevertheless, I follow a technique developed by Altonji et al. (2005) that attempts to obtain information on the degree of selection on unobservables based on the degree of selection on observables to evaluate the selection bias on the estimates of the effects of single-sex schooling.²³

Table 10 reports the results from the propensity score analysis using kernel and five nearest-neighbor matching techniques. As expected given the small and few differences in

 $^{^{22}}$ I use the Stata command *psmatch2* (see Leuven and Sianesi, 2003) to calculate the propensity score estimates.

 $^{^{23}}$ See Altonji et al. (2005) for a detailed explanation of the technique.

student characteristics and the large overlap in estimated propensity scores (not shown), the point estimates are quite similar to the OLS estimates for both the conventional student background control set (see Table 1) and the extensive student background control set (see Tables 1 and 2). Overall, I find positive, significant effects for girls at single-sex schools, but no effects for boys in either OLS or propensity score analysis. The underlying crucial assumption is that after conditioning on observable characteristics, students do not differ in unobservable characteristics. The right-hand side of Table 10 therefore reports the OLS estimates along with the estimated selection bias due to unobservables. In practice, the relationship between single-sex schooling and the observable determinants of math achievement (individual background variables) is used to approximate the relationship between single-sex schooling and influential unobservable factors. The selection bias is then estimated based on the underlying assumption that the selection on observables and the selection on unobservables are of equal magnitude. Since observable background information, such as parents' education and number of books at home, is collected to reduce potential bias, the relationship between unobservable characteristics of student achievement and single-sex schooling is likely to be even less strong. Interestingly, the estimated selection bias is negative and ranges from -0.012 to -0.071, depending on the set of controls. The bias is larger for the set of controls that includes exclusively strong predictors of student achievement and is close to zero when a large number of background controls is added. Given these results, a substantial positive single-sex school effect for girls cannot be rejected. The OLS estimates even provide a lower bound of the single-sex schooling effect given the negative sign of the selection bias. The reported selection bias for boys is also negative and quite large, meaning that, I cannot rule out that there is a positive effect of single-sex schooling for boys, too.

7 Discussion and Conclusion

Empirical results on the effects of single-sex schooling are often inconclusive and do not account for potential selection issues. Recently, several studies have addressed these problems and attempt to pinpoint the causal effects of single-sex schooling. Nevertheless, it is necessary to understand the underlying mechanisms and channels of such effects before any policy recommendations can be offered.

This paper contributes to the growing quasi-experimental literature and investigates the effect of single-sex schooling in a particularly interesting setting. In the Korean education system, students are randomly assigned to secondary schools, which can be either single-sex or coeducational. Given that attendance at single-sex schools is orthogonal to student characteristics such as socioeconomic background and ability, the comparison between girls (boys) at coeducational schools and girls (boys) at single-sex schools should identify a reliable effect of single-sex schooling on student achievement. Although there may be confounding factors, several robustness checks suggest that the effects are not driven by

observable and unobservable differences in the types of students who attend single-sex and coeducational schools. Moreover, the rich data-set I use allows me to investigate a large set of potential channels and features that are often associated with single-sex schooling in the public debate. Although arguments for and against single-sex education are well-developed, most empirical work stops after obtaining the reduced-form estimates of the effect of single-sex schooling.

I find substantial positive, significant effects of single-sex schooling for girls from low parental support backgrounds in Math, but no effects for boys. Differences in school and teacher characteristics, gender-tailored education practices, or reduced gender stereotypes at single-sex schools cannot explain the finding. Comparisons across gender reveal that the test score gender gap in math is especially large for girls from low parental support backgrounds who attend coeducational schools. This result suggests that these girls might be somehow harmed by the presence of boys when learning a stereotypically male subject such as math. Given that most Western countries report large gender test score gaps in math while educating their students in coeducational schools, this is an particular interesting finding (see also Guiso et al., 2008; Fryer and Levitt, 2010).

In this regard, it must be remembered that schooling tradition and culture in Korea obviously differs from that of Western societies. Moreover, the data I analyze relate to a point of time when gender equality levels, as measured by, for example, the gender gap index (GGI), were relatively low in Korea. Even though this raises concerns about the generalizability of the findings, this paper documents an interesting pattern that is consistent with earlier findings. The fact that the in-depth analysis cannot fully explain the positive effects for girls suggests that future research should focus more on classroom interactions when trying to understand the underlying mechanisms of the effects of single-sex schooling.

- Akerlof, G. A. and Kranton, R. E. (2000). Economics and identity. The Quarterly Journal of Economics, 115 (3):715–753.
- Altonji, J. G., Elder, T. E., and Taber, C. R. (2005). Selection on observed and unobserved variables: Assessing the effectiveness of catholic schools. *Journal of Political Economy*, 113 (1):151–184.
- Behrman, J. R., Choi, J., and Park, H. (2012). Causal effects of single-sex schools on college entrance exams and college attendance: Random assignment in seoul high schools. *Demography*, forthcoming.
- Beyer, S. and Bowden, E. M. (1997). Gender Differences in Self-Perceptions: Convergent Evidence from Three Measures of Accuracy and Bias. *Personality and Social Psychology Bulletin*, 23:157 – 172.
- Bigler, R. S. and Signorella, M. L. (2011). Single-sex education: New perspectives and evidence on a continuing controversy. *Sex Roles*, 65:659–669.
- Billger, S. M. (2009). On reconstructing school segregation: The efficacy and equity of single-sex schooling. *Economics of Education Review*, 28:393–402.
- Booth, A. L., Cardona, L., and Nolen, P. J. (2011). Gender differences in risk aversion: Do single-sex environments affect their development? Technical report, IZA Discussion Paper No. 6133.
- Cohen, A. (2012). Ew, boys: The brewing legal battle over same-sex education.
- Coleman, J. S. (1961). The Adolescent Society. Free Press of Glencoe.
- Eisenkopf, G., Hessami, Z., Fischbacher, U., and Ursprung, H. (2011). Academic performance and single-sex schooling: Evidence from a natural experiment in switzerland. Technical report, CESIfo Working Paper No. 3592.
- Favara, M. (2011). The cost of acting "girly": Gender stereotypes and educational choices. Technical report, Institute for Social and Economic Research (ISER).
- Fryer, R. G. and Levitt, S. D. (2010). An empirical analysis of the gender gap in mathematics. *American Economic Journal: Applied Economics*, 2 (2):210–240.
- Guiso, L., Monte, F., Sapienza, P., and Zingales, L. (2008). Culture, gender, and math. Science, 320(5880):1164–65.
- Halpern, D. F., Eliot, L., Bigler, R. S., Fabes, R. A., Hanish, L. D., Hyde, J., Liben, L. S., and Martin, C. L. (2011). The pseudoscience of single-sex schooling. *Science* 23, 333 (6050):1706–1707.

- Hanushek, E. A. and Woessmann, L. (2011). The Economics of International Differences in Educational Achievement. In Eric A. Hanushek, S. M. and Woessmann, L., editors, *Handbooks in Economics*, volume 3, pages 89–200, The Netherlands: North-Holland.
- Heckman, J. J. (2006). Skill formation and the economics of investing in disadvantaged children. Science, 312:1900–1901.
- Hoxby, C. (2000). Peer effects in the classroom: Learning from gender and race variation. Working Paper 7867, National Bureau of Economic Research.
- Jackson, C. (2002). Can single-sex classes in co-educational schools enhance the learning experiences of girls and/or boys? an exploration of pupils' perceptions. British Educational Research Journal, 28(1):37–48.
- Jackson, C. K. (2012). Single-sex schools, student achievement, and course selection: Evidence from rule-based student assignments in trinidad and tobago. *Journal of Public Economics*, 96:173–187.
- Jimenez, E. and Lockheed, M. E. (1989). Enhancing girls' learning through single-sex education: Evidence and a policy conundrum. *Educational Evaluation and Policy Analysis*, 11 (2):117–142.
- Joshi, H., Leonard, D., and Sullivan, A. (2010). Single-sex schooling and academic attainment at school and through the lifecourse. *American Educational Rsearch Journal*, 47 (1):6–36.
- Jürges, H. and Schneider, K. (2010). Central exit examinations increase performance...but take the fun out of mathematics. *Journal of Population Economics*, 23(2):497–517.
- Kim, S. and Lee, J.-H. (2003). The secondary school equalization policy in south korea. Technical report, KDI SchoolWorkingPaper 02-05.
- Kim, T., Lee, J.-H., and Lee, Y. (2008). Mixing versus sorting in schooling: Evidence from the equalization policy in south korea. *Economics of Education Review*, 27(6):697 711.
- Lavy, V. and Schlosser, A. (2011). Mechanisms and impacts of gender peer effects at school. American Economic Journal: Applied Economics, American Economic Association, 3 (2):1–33.
- Lavy, V., Silva, O., and Weinhardt, F. (2012). The Good, the Bad and the Average: Evidence on the Scale and Nature of Ability Peer Effects in Schools. *Journal of Labor Economics*, 30(2):367 – 414.
- Lee, C. J. (2004). The korean equalization trap. Technical report.

- Lee, V. E. and Bryk, A. S. (1986). Effects of single-sex secondary schools on student achievement and attitudes. *Journal of Educational Psychology*, 78:381–395.
- Leuven, E. and Sianesi, B. (2003). Psmatch2: Stata module to perform full mahalanobis and propensity score matching, common support graphing, and covariate imbalance testing. Technical report, Boston College Department of Economics, Statistical Software Components.
- Marsh, H. W. (1989). Effects of attending single-sex and coeducational high schools on achievement, attitudes, behaviors, and sex differences. *Journal of Educational Psychology*, 81:70–85.
- Neidorf, T. S., Binkley, M., Gattis, K., and Nohara, D. (2006). Comparing mathematics content in the national assessment of educational progress (naep), trends in international mathematics and science study (timss), and program for international student assessment (pisa) 2003 assessments. Technical report, Washington: National Center for Education Statistics (May).
- OECD (2009). Pisa 2009 results: What students know and can do student performance in math, science and reading.
- Paglin, M. and Rufolo, A. M. (1990). Heterogeneous human capital, occupational choice, and male-female earnings differences. *Journal of Labor Economics*, 8 (1):123–144.
- Riordan, C. (1990). *Girls and Boys in School: Together or Separate?* New York: Teachers College Press.
- Schütz, G., Ursprung, H. W., and Wössmann, L. (2008). Education policy and equality of opportunity. *Kyklos*, 61:279 – 308.
- Thompson, J. S. (2003). The effect of single-sex secondary schooling on women's choice of college major. *Sociological Perspectives*, 46 (2):257–278.
- Whitmore, D. (2005). Resource and Peer Impacts on Girls' Academic Achievement: Evidence from a Randomized Experiment. *The American Economic Review*, 95(2):199 – 203. Papers and Proceedings of the OneHundred Seventeenth Annual Meeting of the American Economic Association, Philadelphia, PA, January 7-9, 2005 (May, 2005).
- Woessmann, L. (2003). Schooling Resources, Educational Institutions and Student Performance: The International Evidence. Oxford Bulletin of Economics & Statistics, 65(2):117 – 170.
- Woessmann, L. (2008). How Equal are Educational Opportunities? Family Background and Student Achievement in Europe and the United States. *Zeitschrift für Betriebswirtschaft*, 78 (1):45 – 70.

Woessmann, L. and Piopiunik, M. (2009). Was unzureichende bildung kostet: Eine berechnung der folgekosten durch entgangenes wirtschaftswachstum. Technical report, Bertelsmann Stiftung.

Figures and Tables



Figure 1. Student Population by School Type and Gender.

Notes: The white squares denote the share of girls and boys at each type of school. The bar denotes the share of girls and boys attending single-sex schools.

Table 1	
Student	Characteristics

		Female			Male	
	Coed	SS	Diff.	Coed	SS	Diff.
Age	14.45	14.42	-0.03 **	14.42	14.43	0.01
-	(0.32)	(0.33)	(0.01)	(0.34)	(0.35)	(0.01)
None	0.04	0.04	0.00	0.03	0.05	0.02**
	(0.19)	(0.20)	(0.01)	(0.17)	(0.21)	(0.01)
Primary	0.13	0.13	0.00	0.12	0.13	0.02
	(0.34)	(0.34)	(0.01)	(0.32)	(0.34)	(0.01)
Secondary	0.47	0.55	0.08 ***	0.42	0.47	0.05**
	(0.50)	(0.50)	(0.02)	(0.49)	(0.50)	(0.02)
University	0.29	0.22	-0.07 ***	0.35	0.25	-0.10***
	(0.45)	(0.41)	(0.02)	(0.48)	(0.43)	(0.02)
0-10 Books	0.08	0.08	-0.00	0.08	0.11	0.03**
	(0.27)	(0.27)	(0.01)	(0.28)	(0.31)	(0.01)
11-25 Books	0.11	0.09	-0.02	0.09	0.11	0.02
	(0.31)	(0.29)	(0.01)	(0.29)	(0.31)	(0.01)
26-100 Books	0.35	0.37	0.02	0.35	0.35	-0.00
	(0.48)	(0.48)	(0.02)	(0.48)	(0.48)	(0.02)
101-200 Books	0.25	0.24	-0.00	0.24	0.22	-0.01
	(0.43)	(0.43)	(0.02)	(0.42)	(0.42)	(0.02)
> 200 Books	0.21	0.21	0.01	0.24	0.21	-0.03*
	(0.41)	(0.41)	(0.02)	(0.42)	(0.41)	(0.02)
Live w Parents	0.90	0.90	-0.00	0.90	0.90	-0.01
	(0.30)	(0.30)	(0.01)	(0.30)	(0.30)	(0.01)
Observations	1101	1247	1056	413	643	1056

 $\it Note:$ Individual observations are weighted by sampling probabilities. Standard deviations in parentheses. Data source: TIMSS 1999.

		Female			Male	
	Coed	\mathbf{SS}	Diff.	Coed	\mathbf{SS}	Diff.
Home resources						
Computer at home	0.63	0.63	0.00	0.77	0.68	-0.09***
	(0.48)	(0.48)	(0.02)	(0.42)	(0.47)	(0.02)
Observation	1099	1246	2345	1073	1351	2424
Internet at home	0.20	0.22	0.01	0.33	0.23	-0.10***
	(0.40)	(0.41)	(0.02)	(0.47)	(0.42)	(0.02)
Observation	1076	1207	2283	1049	1314	2363
Calculator at home	0.95	0.96	0.02^{*}	0.97	0.96	-0.00
	(0.22)	(0.19)	(0.01)	(0.18)	(0.18)	(0.01)
Observation	1098	1246	2344	1072	1350	2422
Read a book						
about every day	0.22	0.22	0.00	0.28	0.25	-0.03
	(0.42)	(0.42)	(0.02)	(0.45)	(0.43)	(0.02)
about once a week	0.44	0.43	-0.01	0.38	0.38	0.00
	(0.50)	(0.50)	(0.02)	(0.49)	(0.49)	(0.02)
rarely/once a month	0.34	0.35	0.01	0.34	0.37	0.03
	(0.47)	(0.48)	(0.02)	(0.48)	(0.48)	(0.02)
Observation	1096	1244	2340	1063	1342	2405
Watch news or documentaries						
about every day	0.27	0.31	0.04^{**}	0.30	0.30	0.00
	(0.44)	(0.46)	(0.02)	(0.46)	(0.46)	(0.02)
about once a week	0.33	0.35	0.03	0.34	0.36	0.02
	(0.47)	(0.48)	(0.02)	(0.47)	(0.48)	(0.02)
rarely/once a month	0.41	0.34	-0.06***	0.37	0.34	-0.03
	(0.49)	(0.47)	(0.02)	(0.48)	(0.47)	(0.02)
Observation	1093	1244	2337	1057	1335	2392
Go to the movies						
about every day/ once a week	0.03	0.03	0.00	0.04	0.06	0.02^{*}
	(0.17)	(0.18)	(0.01)	(0.20)	(0.23)	(0.01)
about once a month	0.46	0.46	0.00	0.45	0.44	-0.01
	(0.50)	(0.50)	(0.02)	(0.50)	(0.50)	(0.02)
rarely	0.51	0.50	-0.01	0.51	0.50	-0.01
	(0.50)	(0.50)	(0.02)	(0.50)	(0.50)	(0.02)
Observation	1098	1244	2342	1063	1337	2400
Watch opera, ballet, classic music						
about every day/ once a week	0.06	0.06	-0.00	0.01	0.01	-0.01
	(0.24)	(0.24)	(0.01)	(0.12)	(0.09)	(0.00)
about once a month	0.18	0.17	-0.00	0.12	0.14	0.03^{**}
	(0.38)	(0.38)	(0.02)	(0.32)	(0.35)	(0.01)
rarely	0.76	0.76	0.00	0.83	0.80	-0.03*
	(0.43)	(0.42)	(0.02)	(0.37)	(0.40)	(0.02)
Observation	1091	1245	2336	1055	1333	2388
Watch comedies						
about every day	0.53	0.52	-0.01	0.54	0.54	-0.01
	(0.50)	(0.50)	(0.02)	(0.50)	(0.50)	(0.02)
about once a week	0.34	0.35	0.01	0.34	0.34	0.00
	(0.48)	(0.48)	(0.02)	(0.47)	(0.47)	(0.02)
rarely/about once a month	0.13	0.13	0.00	0.07	0.07	0.00
	(0.33)	(0.34)	(0.01)	(0.25)	(0.26)	(0.01)
Observation	1097	1242	2339	1062	1338	2400

 $\it Note:$ Individual observations are weighted by sampling probabilities. Standard deviations in parentheses. Data source: TIMSS 1999.

				Fema	le			
		Ma	ath			Scie	ence	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Single-Sex	0.127^{**} (0.058)	$\begin{array}{c} 0.144^{***} \\ (0.049) \end{array}$	$\begin{array}{c} 0.118^{**} \\ (0.052) \end{array}$	$\begin{array}{c} 0.135^{***} \\ (0.049) \end{array}$	$0.062 \\ (0.057)$	$\begin{array}{c} 0.070 \\ (0.053) \end{array}$	$0.059 \\ (0.054)$	$\begin{array}{c} 0.070 \\ (0.054) \end{array}$
Student Controls School Controls Teacher Controls Imputation Controls	No No No No	Yes No No Yes	Yes Yes No Yes	Yes Yes Yes Yes	No No No No	Yes No No Yes	Yes Yes No Yes	Yes Yes Yes
Observations Cluster R^2	$2348 \\ 76 \\ 0.004$	$2348 \\ 76 \\ 0.173$	$2348 \\ 76 \\ 0.182$	$2348 \\ 76 \\ 0.189$	$2348 \\ 76 \\ 0.001$	$2348 \\ 76 \\ 0.137$	$2348 \\ 76 \\ 0.148$	$2348 \\ 76 \\ 0.157$
				Male	e			
		Ma	ath			Scie	ence	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Single-Sex	-0.087 (0.065)	-0.016 (0.050)	-0.002 (0.050)	-0.001 (0.058)	-0.092 (0.068)	-0.028 (0.054)	-0.030 (0.052)	-0.042 (0.062)
Student Controls School Controls Teacher Controls Imputation Controls	No No No	Yes No No Yes	Yes Yes No Yes	Yes Yes Yes	No No No No	Yes No No Yes	Yes Yes No Yes	Yes Yes Yes
Observations Cluster	2427 78	2427 78	2427 78	2427 78	2427 78	2427 78	2427 78	2427 78
R ² Data	0.002	0.176 TIMS	0.186 S 1999	0.199	0.002	0.167 TIMS	0.175 S 1999	0.182

Table 3Effects of Single-Sex Education at Middle Schools

Notes: Individual student observations are weighted by sampling probabilities. Standard errors are clustered at the school level and reported in parentheses. Student controls include age, parent's education, books at home and living with mother and father. School controls include total enrollment, school location, student-teacher- and computer-student ratios, hiring and course autonomy. Teacher controls include teacher's age, gender, education and books at home for the teacher reported first if there are several. The regressions control for the fact that some students have several teachers in math and science. All regressions control for imputation. * p<0.10, ** p<0.05, *** p<0.01

			Female Math	Test Score		
		Socioeconomic Background		University Deg	gree	Peer Pressure
	Few Books	High Home Resources	Family Size	At least one Parent	Mother	Study Time
	(1)	(2)	(3)	(4)	(5)	(9)
Single-Sex	0.142^{***}	0.140^{***}	0.142^{***}	0.149^{***}	0.163^{***}	0.151^{***}
D	(0.049)	(0.051)	(0.049)	(0.051)	(0.051)	(0.043)
Peers' Average	-0.176	0.126	-0.082	0.141	0.469^{*}	0.732^{***}
	(0.214)	(0.218)	(0.062)	(0.168)	(0.269)	(0.267)
Full Controls	$\mathbf{Y}_{\mathbf{es}}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Imputation Controls	Yes	Yes	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}
Observations	2348	2346	2336	2348	2348	2281
Cluster	26	92	26	26	76	26
R^2	0.189	0.189	0.190	0.188	0.185	0.190
			Male Math 7	Test Score		
		Socioeconomic Background		University Deg	gree	Peer Pressure
	Few Books	High Home Resources	Family Size	At least one Parent	Mother	Study Time
	(1)	(2)	(3)	(4)	(5)	(9)
Single-Sex	0.018	0.016	-0.021	0.012	0.008	0.027
D	(0.056)	(0.058)	(0.057)	(0.057)	(0.056)	(0.060)
Peers' Average	-0.463^{***}	0.250	-0.082	0.232^{*}	0.382^{**}	0.513
	(0.147)	(0.170)	(0.062)	(0.138)	(0.168)	(0.367)
Full Controls	Yes	${ m Yes}$	Yes	${ m Yes}$	m Yes	m Yes
Imputation Controls	Yes	Yes	Yes	Yes	\mathbf{Yes}	\mathbf{Yes}
Observations	2427	2425	2397	2427	2427	2349
Cluster	78	78	78	78	78	78
R^2	0.202	0.201	0.200	0.200	0.201	0.196
Notes: Individual studen controls include age, paren and computer-student rati if there are several). The $1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 + 1 $	tt observations are v nt's education, book ios, hiring and cour regressions control f	veighted by sampling probabilitie is at home and living with moth se autonomy. Teacher controls in or the fact that some students ha	s. Standard errors ar er and father. School iclude teacher's age, g ave several teachers in	e clustered at the school level l controls include total enroll gender, education and books 1 math and science. All regree	l and reported in p nent, school locati at home (for the te ssions control for in	arentheses. Student on, student-teacher- acher reported first aputation.* p<0.10,

Table 4Effects of Single-Sex Education accounting for Peer Quality

				Math Te	est Score			
	Copying	5 Notes	Having a [Test/ Quiz	Giving He	omework	Working i	n Groups
	Female	Male	Female	Male	Female	Male	Female	Male
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Single-Sex	0.134^{***}	0.015	0.151^{***}	0.010	0.131^{***}	0.007	0.137^{***}	0.004
)	(0.048)	(0.059)	(0.049)	(0.057)	(0.048)	(0.057)	(0.048)	(0.059)
Once in a while	0.095	0.033	0.213^{***}	0.406^{***}	0.191^{**}	0.195^{**}	-0.091^{**}	-0.015
	(0.092)	(0.090)	(0.064)	(0.059)	(0.087)	(0.082)	(0.043)	(0.043)
Pretty often/ Always	0.100	-0.046	0.225^{***}	0.365^{***}	0.223^{**}	0.104	0.051	0.034
	(0.088)	(0.095)	(0.082)	(0.067)	(0.088)	(0.089)	(0.062)	(0.064)
Eull Controls	$\mathbf{V}_{\mathbf{cc}}$	$\mathbf{V}_{\mathbf{rc}}$	V_{00}	\mathbf{V}_{00}	\mathbf{V}_{00}	Voc	$\mathbf{V}_{\mathbf{cc}}$	V_{20}
	1 CD	B 1	1 CD	T CO	TCD	1 CD	T CD	SD I
Imputation Controls	\mathbf{Yes}	\mathbf{Yes}	${ m Yes}$	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	Yes	\mathbf{Yes}
Observations	2348	2427	2348	2427	2348	2427	2348	2427
Cluster	76	78	76	78	76	78	26	78
R^2	0.190	0.199	0.195	0.218	0.191	0.200	0.191	0.197
<i>Notes:</i> Individual studen controls include age, paren and computer-student rati if there are several). The r ** $p<0.05$, *** $p<0.01$	t observations are it's education, boc os, hiring and cou egressions control	weighted by sam oks at home and l rse autonomy. Te for the fact that	pling probabilities. living with mother acher controls incl some students hav	Standard errors a t and father. Schoo lude teacher's age, e several teachers	re clustered at the ol controls include gender, education in math and science	school level and : total enrollment, and books at hor e. All regressions	reported in parentl school location, st ne (for the teacher control for imputa	heses. Student udent-teacher- reported first tion.* p<0.10,

Table 5Effects of Single-Sex Education accounting for Teaching Practices

			Female Math Test Score		
	Like Math	Positive Attitude	Important for myself	Confidence	Educational Aspirations
	(1)	(2)	(3)	(4)	(5)
Single-Sex	0.132^{***}	0.112^{**}	0.112^{**}	0.142^{***}	0.130^{***}
Medium	(0.044) 0.598^{***}	(0.046) 0.363^{***}	(0.048) 0.297^{***}	(0.046) 0.723^{***}	(0.044) 0.504^{***}
	(0.040)	(0.045)	(0.081)	(0.078)	(0.074)
High	0.974^{***}	0.924^{***} (0.060)	0.529^{***} (0.082)	1.377^{***} (0.097)	1.032^{***}
Full Controls Immitation Controls	Yes Voc	${ m Yes}_{ m voc}$	${ m Yes}_{ m Voc}$	${ m Yes}_{ m Voc}$	${ m Yes}_{ m voc}$
	109	IGS	IGS	IGS	169
Observations	2342	2339	2341	2342	2342
Cluster	26	76	76	26	26
R^2	0.305	0.247	0.211	0.251	0.286
			Male Math Test Score		
	Like Math	Positive Attitude	Important for myself	Confidence	Educational Aspirations
	(1)	(2)	(3)	(4)	(5)
Single-Sex	0.014	0.013	0.014	0.016	0.011
5	(0.048)	(0.055)	(0.054)	(0.052)	(0.052)
Medium	0.615^{***}	0.322^{***}	0.388***	0.586^{***}	0.410^{***}
	(0.037)	(0.042)	(0.063)	(0.087)	(0.084)
High	0.838*** (0.053)	0.731*** (0.079)	0.536*** (0.061)	1.167*** (0.001)	0.900*** (0.058)
	(0000)	(210.0)	(100.0)	$(\tau e \alpha \cdot \alpha)$	(000.0)
Full Controls	${ m Yes}$	Yes	Yes	\mathbf{Yes}	Yes
Imputation Controls	Yes	Yes	Yes	Yes	Yes
Observations	2414	2413	2416	2412	2416
Cluster	78	78	78	78	78
R^2	0.305	0.237	0.221	0.246	0.282
Notes: Individual student controls include age, parent and computer-student ratio if there are several). The re ** p<0.05, *** p<0.01	observations are weig 's education, books at s, hiring and course a gressions control for t	hted by sampling probabilities t home and living with mothe utonomy. Teacher controls in he fact that some students ha	s. Standard errors are clustered a pr and father. School controls inc clude teacher's age, gender, educ ve several teachers in math and s	at the school level and r clude total enrollment, s ation and books at hom science. All regressions o	sported in parentheses. Student chool location, student-teacher- e (for the teacher reported first control for imputation.* $p < 0.10$

Table 6Effects of Single-Sex Education accounting for Students' Attitudes

				remar	A TANK NEAT TIMPIN A			
				Limits	to Teaching		Behavior pres	ents Problem
I	Students'	Behavior	Stu	dents	Wide]	Range of	Studen	its get
0	Orderly	As told	Disruptive	Uninterested	Backgrounds	Acad. Abilities	Intimidated	Injured
I	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Single-Sex 0).121**	0.134***	0.126**	0.095*	0.142***	0.083*	0.114**	0.077
) Medium	(0.049) 0.018	(0.048) 0.219^{**}	(0c0.0) -0.009	(0.051) -0.051	(160.0) 0.078	(0.048) - 0.200^{***}	(0.052) -0.016	$(0.048) -0.137^{***}$
)	(0.077)	(0.097)	(0.062)	(0.067)	(0.068)	(0.075)	(0.067)	(0.052)
High 0).191** `^^^^	0.211^{**}	0.023	-0.202^{**}	-0.001	-0.087	-0.179^{**}	-0.111
Full Controls	(1.091)	(0.098)	$_{ m Yes}^{(0.082)}$	(0.101) Yes	(0.11()) Yes	$_{ m Yes}^{(0.074)}$	(0.0.6)	(0.07δ)
Imputation Controls	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	Yes	Yes
Observations	2348	2348	2281	2281	2281	2327	2249	2249
Clusters	26	26	74	74	74	75	73	73
R^{2}	0.195	0.191	0.191	0.194	0.192	0.193	0.189	0.190
				Male	Math Test Score			
I				Limits	to Teaching		Behavior pres	ents Problem
1	Students'	Behavior	Stu	dents	Wide]	Range of	Studer	its get
C	Orderly	As told	Disruptive	Uninterested	Backgrounds	Acad. Abilities	Intimidated	Injured
Ι	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Single-Sex	0.004	0.006	-0.022	-0.00	0.016	-0.008	-0.001	-0.008
;	(0.057)	(0.058)	(0.061)	(0.066)	(0.055)	(0.060)	(0.058)	(0.055)
Medium	0.128**	0.145^{**}	0.021 0.066)	-0.043	0.100** (0.059)	0.009	6T0.0-	010.0
High	(0.105)	(0.004) 0.081	-0.098	-0.069 -0.069	(0.030)	-0.064	(0.038) -0.038	(0.00)
	(0.069)	(0.072)	(0.073)	(0.073)	(0.070)	(0.117)	(0.093)	(0.080)
Full Controls	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	Yes
Imputation Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	2427	2427	2401	2401	2401	2401	2411	2411
Clusters	76	76	74	74	74	75	73	73
R^2	0.199	0.199	0.200	0.199	0.200	0.199	0.197	0.198
Reported by	Stud	lents		Te	achers		Princ	ipals

Table 7 Effects of Single-Sex Education accounting for Disciplinary Classroom Climate

			Test Sco	ore Math		
		Female			Male	
	(1)	(2)	(3)	(4)	(5)	(6)
Single-Sex	0.051 (0.059)	0.041 (0.047)	0.093^{*} (0.049)	-0.020 (0.070)	-0.032 (0.066)	$0.019 \\ (0.054)$
Low Socioeconomic	-0.550^{***} (0.058)			-0.435^{***} (0.054)		
Low*Single-Sex	0.177^{**} (0.074)			(0.022) (0.079)		
Low-educated		-0.370***			-0.329***	
Low-educated*Single-Sex		(0.055) 0.230^{***} (0.070)			$(0.059) \\ 0.082 \\ (0.084)$	
Math not important for Mother			-0.708^{***}			-0.780^{***}
NotImp*Single-Sex			(0.177) 0.547^{**} (0.221)			(0.135) 0.071 (0.180)
Full Controls	Yes	Yes	Yes	Yes	Yes	Yes
Imputation Controls	Yes	Yes	Yes	Yes	Yes	Yes
Observations	2348	2348	2335	2427	2427	2406
Cluster R^2	$76 \\ 0.162$	$\begin{array}{c} 76 \\ 0.163 \end{array}$	$\begin{array}{c} 76 \\ 0.195 \end{array}$	$78 \\ 0.162$	$78 \\ 0.176$	$78 \\ 0.224$

Table 8Heterogenous Effects of Single-Sex Education

Notes: Individual student observations are weighted by sampling probabilities. Standard errors are clustered at the school level and reported in parentheses. Student controls include age, parent's education, books at home and living with mother and father. School controls include total enrollment, school location, student-teacher- and computer-student ratios, hiring and course autonomy. Teacher controls include teacher's age, gender, education and books at home (for the teacher reported first if there are several). The regressions control for the fact that some students have several teachers in math. All regressions control for imputation.* p<0.10, ** p<0.05, *** p<0.01

					Math				
				Socioe	economic Backgro	pund			
		All			Low			High	
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)
Female	-0.040	-0.086**	-0.029	-0.071	-0.166^{***}	-0.079	0.012	0.014	0.053
Single-Sex	(+0.00)	(0.040) 0.051	0.039 0.039 0.059)	(0.040)	(100.0) 0.063 (10.061)	(0.053)	(000.0)	(0.040) (0.040)	(0.004) 0.028 (0.070)
Single-Sex*Female		(0.041)	(0.032)		0.156*	$\begin{array}{c} (0.009) \\ 0.131 \end{array}$		(000.0) (000-	-0.020
injured2		(0.066)	(0.070) 0.036		(0.086)	(0.089) 0.021		(0.079)	(0.084) 0.053
iniured3			(0.049) 0.045			(0.061) 0.074			(0.069) 0.055
			(0.062)			(0.091)			(0.083)
$injured2^*Female$			-0.125*(0.067)			-0.166^{*}			-0.096
$injured3^*Female$			-0.066 (0.074)			-0.170 (0.106)			-0.009 (0.111)
Full Controls Imputation Control	$\mathop{\rm Yes}\limits_{\mathop{\rm Yes}}$	$\substack{\text{Yes}}{\text{Yes}}$	${ m Yes}{ m Yes}$	${ m Yes}$	$\mathop{\rm Yes}\limits_{\mathop{\rm Yes}}$	$\substack{\text{Yes}}{\text{Yes}}$	Yes Yes	Yes Yes	${ m Yes}$
Observations	4775	4775	4660	2575	2575	2500	2200	2200	2160
Cluster	116	116	113	116	116	113	116	116	113
R^2	0.181	0.183	0.182	0.124	0.129	0.130	0.110	0.110	0.112
Notes: Individual studen controls include age, pare and computer-student rat if there are several). The ** $p<0.05$, *** $p<0.01$	nt observations nt's education ios, hiring and regressions cor	are weighted by , books at home I course autonom itrol for the fact	sampling proba and living with y. Teacher conti that some stude	bilities. Standar mother and fath rols include teach nts have several	d errors are cluste ter. School contro ner's age, gender, teachers in math	red at the schoc old include total education and b and science. All	ol level and repo enrollment, scho ooks at home (; regressions cont	rted in parenthe ool location, stu for the teacher 1 trol for imputati	ses. Student dent-teacher- eported first ion.* $p<0.10$,

Table 9Single-Sex Education and the Test Score Gap in Math

-31 -

				Fen	nale			
		Propensity Sco	ore Matching			Bias in	1 OLS	
	Kei	rnel	Near.N	leighbor	OLS Es	stimate	Estimat	ted Bias
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Single-Sex	0.117^{***} (0.042)	0.107^{**} (0.043)	0.086 (0.054)	0.104^{**} (0.046)	0.143^{***} (0.049)	0.125^{**} (0.048)	-0.071 (0.025)	-0.012 (0.020)
Student Background Variables								
Set 1 Set 2	${ m Yes}$ No	${ m Yes}{ m Yes}$	${ m Yes}$ No	m Yes $ m Yes$	${ m Yes}$ No	${ m Yes}{ m Yes}$	${ m Yes}$ No	Yes Yes
Observations LR-chi2	2348 37.71	2260 59.69	2348 37.71	2260 59.69	2348 0.171	2260 0.212	2348 0.171	2260 0.212
)			M.	ale			
		Propensity Sco	ore Matching			Bias in	I OLS	
	Kei	rnel	Near.N	leighbor	OLS Es	stimate	Estimat	ted Bias
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Single-Sex	-0.033	0.007	-0.023	0.001	-0.016	0.014	-0.372	-0.342
Student Background Variables	(0.044)	(0+0-0)	(0.040)	(0.040)	(0000)	(1.0.04)	(000.0)	(000.0)
Set 1	\mathbf{Yes}	Yes	\mathbf{Yes}	\mathbf{Yes}	\mathbf{Yes}	Yes	Yes	\mathbf{Yes}
Set 2	No	\mathbf{Yes}	No	\mathbf{Yes}	N_{O}	\mathbf{Yes}	N_{O}	Yes
Observations	2427	2310	2427	2310	2427	2310	2427	2310
LR-chi2	47.80	2310	47.80	91.92	0.164	0.238	0.164	0.238
Notes: This table reports propensit Student control set 1 includes age, m internet at home, calculator at home frequency of watching opera, ballet.	ty score matching to there and father a, frequency of re- frequency of wat	g results and ordi rs education, bool ading a book at 1 sching comedies.	nary least squan ss at home and l nome, frequency Standard deviat	re results along v living with moth of watching new	with the estimated er and father. Stu vs or documentari see. Data source:	l selection bias l dent control set es at home, freq TIMSS 1999	based on Altonji 2 includes comp uency of going t	et al. (2005). uter at home, o the movies,

Table 10Robustness: Matching Estimates and Selection Bias

Historical Background

As a response to very low enrollment rates after the Japanese liberalization, primary schooling in Korea became universal and compulsory in 1951. Although school facilities and resources were limited after the 3-year Korean War, enrollment rates for elementary schooling increased remarkably and rose steadily (Kim and Lee, 2003). Since most resources were invested in the primary education sector, the capacity of public secondary schools was not much increased. As a result, the provision of secondary school facilities lagged behind the rapid growth of the student population and resulted in a fierce competition among students in the admission process to middle and high schools. Consequently, all middle and high schools selected their students through competitive entrance examinations. However, the selection of students based on entrance examinations resulted in an advantage for wealthy families that were able to better support their children, particularly by paying for private tutoring. At that time, the Korean education system was characterized by an excess demand for secondary schools, substantial quality differences across schools, and overall unequal education opportunities. As a response, an "Equalization Policy" (EP) was introduced in 1969 with the aim of creating equal education opportunities at middle schools and reducing the influence of social background on student educational achievement.

		Middle	Schools	
	Coed		Single-Sex Schools	3
	All	All	Female	Male
Total enrollment	1317.98	1178.80	1204.79	1155.09
	(482.17)	(338.10)	(310.60)	(359.84)
Outskirts of a city	0.52	0.34	0.33	0.34
	(0.50)	(0.47)	(0.47)	(0.47)
Center of a city	0.48	0.66	0.67	0.66
	(0.50)	(0.47)	(0.47)	(0.47)
Student-Teacher-Ratio	25.35	24.79	24.55	25.00
	(5.67)	(4.06)	(3.06)	(4.78)
Student-Computer-Ratio	39.25	61.92	49.68	73.08
	(37.92)	(152.53)	(54.68)	(203.74)
Share Teacher > 5 years	14.15	22.88	26.59	19.51
	(26.02)	(34.36)	(38.06)	(30.21)
Hiring Autonomy	0.27	0.42	0.46	0.40
	(0.42)	(0.49)	(0.49)	(0.49)
Course Autonomy	0.92	0.94	0.95	0.94
	(0.26)	(0.24)	(0.22)	(0.24)
Low socioeconomic background	0.53	0.55	0.54	0.57
	(0.50)	(0.50)	(0.50)	(0.50)
Female Teacher	0.64	0.60	0.79	0.42
	(0.48)	(0.49)	(0.41)	(0.49)
Teaching Experience	12.35	13.17	10.64	15.48
	(8.61)	(9.33)	(7.66)	(10.08)
Under 30	0.292	0.129	0.160	0.102
	(0.455)	(0.336)	(0.367)	(0.302)
30-50	0.620	0.695	0.782	0.616
	(0.486)	(0.460)	(0.413)	(0.486)
> 50	0.088	0.175	0.058	0.282
	(0.283)	(0.380)	(0.235)	(0.450)
Master/Phd	0.14	0.12	0.11	0.13
,	(0.35)	(0.32)	(0.31)	(0.33)
Up to 1 bookcase	0.256	0.378	0.311	0.439
1	(0.436)	(0.485)	(0.463)	(0.496)
2 bookcases	0.269	0.308	0.349	0.272
	(0.444)	(0.462)	(0.477)	(0.445)
3 bookcases	0.475	0.314	0.341	0.289
	(0.499)	(0.464)	(0.474)	(0.453)
Ν	2175.00	2600.00	1247.00	1353.00

Table A.1School Characteristics Middle Schools

Note: TIMSS 1999. Individual observations weighted by sampling probabilities. Standard deviations in parentheses. Students with less than 100 books at home are classified as students with low socioeconomic background.

	Fei	male	М	ale
	Math	Science	Math	Science
	(1)	(2)	(3)	(4)
Single-Sex	0.135***	0.062	0.004	-0.017
	(0.049)	(0.050)	(0.058)	(0.053)
Age	0.048	0.157^{***}	0.010	0.067
	(0.048)	(0.054)	(0.051)	(0.060)
None	0.309^{**}	0.249	0.515^{***}	0.545^{***}
	(0.146)	(0.150)	(0.121)	(0.096)
Primary	0.338^{***}	0.398^{***}	0.379^{***}	0.310^{***}
	(0.111)	(0.108)	(0.084)	(0.075)
Secondary	0.379^{***}	0.429^{***}	0.446^{***}	0.422^{***}
	(0.096)	(0.097)	(0.074)	(0.069)
University	0.756^{***}	0.684^{***}	0.698^{***}	0.636^{***}
	(0.100)	(0.100)	(0.076)	(0.074)
11-25 books	0.343^{***}	0.256^{***}	0.237^{**}	0.206^{**}
	(0.107)	(0.088)	(0.095)	(0.083)
26-100 books	0.607^{***}	0.521^{***}	0.602^{***}	0.508^{***}
	(0.092)	(0.092)	(0.067)	(0.049)
101-200 books	0.858^{***}	0.731^{***}	0.756^{***}	0.678^{***}
	(0.096)	(0.094)	(0.071)	(0.060)
> 200 books	1.037^{***}	1.007^{***}	0.985^{***}	0.939^{***}
	(0.088)	(0.085)	(0.078)	(0.061)
Live w parents	0.138^{*}	-0.064	0.205^{***}	0.095
	(0.071)	(0.065)	(0.063)	(0.062)
Total enrollment	0.000	-0.000	-0.000	-0.000
	(0.000)	(0.000)	(0.000)	(0.000)
Center of a city	0.077	0.046	0.077	0.114^{**}
	(0.056)	(0.054)	(0.060)	(0.054)
Student-teacher-ratio	-0.014*	-0.011	-0.004	0.009
	(0.008)	(0.008)	(0.006)	(0.006)
Student-computer-ratio	-0.001**	-0.000	0.000	-0.000
	(0.001)	(0.000)	(0.000)	(0.000)
Share teacher > 5 years	-0.001	-0.002**	-0.001	-0.001
	(0.001)	(0.001)	(0.001)	(0.001)
Autonomy to hire	0.102	0.001	0.008	-0.049
	(0.063)	(0.050)	(0.047)	(0.059)
Autonomy over courses	-0.044	0.057	0.262^{***}	0.247^{**}
	(0.104)	(0.095)	(0.080)	(0.105)
Female teacher	-0.108	-0.009	0.019	0.065
	(0.075)	(0.063)	(0.066)	(0.060)
Teacher age: 30-50	0.032	0.291^{***}	-0.145	-0.070
	(0.082)	(0.073)	(0.094)	(0.103)
Teacher age: > 50	-0.340	0.628^{***}	-0.438**	-0.130
	(0.215)	(0.170)	(0.186)	(0.222)
Master/ Phd	-0.037	0.088	0.115	0.036
	(0.099)	(0.055)	(0.116)	(0.067)
Teaching experience	-0.000	-0.022***	0.007	0.004
	(0.006)	(0.005)	(0.006)	(0.007)
		Continue	d on next page	

Table A.2Effects of Single-Sex Education at Middle Schools

		– continued from p	revious page	
	Fei	male	Ν	fale
	Math	Science	Math	Science
Teacher books: 2 bookcases	0.013	-0.135**	-0.060	-0.019
	(0.079)	(0.062)	(0.078)	(0.066)
Teacher books: 3-4 bookcases	0.060	-0.123*	-0.128	0.009
	(0.062)	(0.064)	(0.081)	(0.053)
Several teacher	0.093	-0.016	-0.320*	-0.028
	(0.120)	(0.057)	(0.173)	(0.066)
Constant	-1.686**	-3.056***	-1.473*	-2.383***
	(0.735)	(0.784)	(0.776)	(0.885)
Imputation control	Yes	Yes	Yes	Yes
Observations	2348	2348	2427	2427
R-sq	0.189	0.157	0.199	0.177

Notes: Individual student observations are weighted by sampling probabilities. Standard errors are clustered at the school level and reported in parentheses. Reference category for parent's education is "do not know", for

		ave Not Urt	derly	Щ	sehave Not as	s Told	Str	idents Disru	ptive	Stud	ents Uninter	ested
	Coed	All-girls	All-Boys	Coed	All-girls	All-Boys	Coed	All-girls	All-Boys	Coed	All-girls	All-Boys
Not a Problem	0.279	0.401	0.274	0.399	0.513	0.421	0.208	0.380	0.238	0.287	0.428	0.368
-	(0.446)	(0.489)	(0.444)	(0.487)	(0.499)	(0.491)	(0.406)	(0.486)	(0.426)	(0.452)	(0.495)	(0.482)
Minor Problem	0.592	0.508	0.594	0.520	0.428	0.492	0.539	0.490	0.568	0.436	0.426	0.470
_	(0.489)	(0.499)	(0.488)	(0.496)	(0.493)	(0.497)	(0.499)	(0.500)	(0.496)	(0.496)	(0.495)	(0.499)
Serious Problem	0.129	0.091	0.132	0.081	0.059	0.086	0.253	0.130	0.194	0.277	0.146	0.162
	(0.333)	(0.286)	(0.336)	(0.271)	(0.235)	(0.279)	(0.435)	(0.337)	(0.396)	(0.448)	(0.353)	(0.369)
Observations	2175	1247	1353	2175	1247	1353	2128	1201	1353	2128	1201	1353
	Wide ra	mge of Back	cgrounds	Wide ra.	nge of Acade.	mic Abilities	Intim	idation of S ¹	tudents	Inj	ury of Stud€	ints
	Coed	All-girls	All-Boys	Coed	All-girls	All-Boys	Coed	All-girls	All-Boys	Coed	All-girls	All-Boys
Not a Problem	0.278	0.508	0.488	0.185	0.480	0.573	0.337	0.362	0.371	0.419	0.698	0.370
	(0.448)	(0.500)	(0.500)	(0.388)	(0.500)	(0.495)	(0.473)	(0.481)	(0.483)	(0.493)	(0.459)	(0.483)
Minor Problem	0.575	0.409	0.435	0.627	0.401	0.355	0.508	0.540	0.536	0.502	0.229	0.507
-	(0.494)	(0.492)	(0.496)	(0.484)	(0.490)	(0.479)	(0.500)	(0.499)	(0.499)	(0.500)	(0.420)	(0.500)
Serious Problem	0.146	0.083	0.078	0.189	0.119	0.071	0.155	0.098	0.093	0.079	0.073	0.123
	(0.354)	(0.276)	(0.268)	(0.391)	(0.324)	(0.258)	(0.362)	(0.298)	(0.290)	(0.270)	(0.260)	(0.328)
Observations	2175	1247	1353	2175	1247	1353	2128	1201	1353	2128	1201	1353

Table A.3Descriptives: Disciplinary Climate

						Teaching	Practice					
	J	Copying Not	es	Hav	ing a Quiz/	Test	Gi	ving Homew	<i>j</i> ork	Wo	orking in Gro	sdno
	Coed	All-girls	All-Boys	Coed	All-girls	All-Boys	Coed	All-girls	All-Boys	Coed	All-girls	All-Boys
Never	0.096	0.057	0.060	0.167	0.174	0.201	0.088	0.039	0.082	0.524	0.417	0.634
	(0.294)	(0.233)	(0.237)	(0.372)	(0.378)	(0.400)	(0.282)	(0.195)	(0.274)	(0.499)	(0.493)	(0.480)
Once in a while	0.426	0.350	0.304	0.572	0.587	0.558	0.451	0.489	0.418	0.312	0.370	0.254
	(0.494)	(0.477)	(0.459)	(0.494)	(0.492)	(0.495)	(0.497)	(0.500)	(0.492)	(0.462)	(0.483)	(0.434)
Pretty often/ always	0.479	0.593	0.636	0.261	0.240	0.241	0.462	0.472	0.500	0.165	0.213	0.112
	(0.499)	(0.491)	(0.481)	(0.439)	(0.427)	(0.427)	(0.498)	(0.499)	(0.499)	(0.371)	(0.409)	(0.314)
Observations	2175	1247	1353	2175	1247	1353	2175	1247	1353	2175	1247	1353
Notes: This table rep	orts descri	ptive statisti	cs on teachin	g practices.	Individual	observations	s are weight	ed by samp	ling probabil	ities. Data	source: TIN	ISS 1999.

Table A.4Descriptives: Teaching Practice

		Lika	Math		d d	Attil Sitiwa Attitud	tude e tourards N	Math		Aath Imnort	ant for mis	
	Fer	nale	M	lale	Fer	nale		lale	Ferr	ale la		fale
	Coed	Single-Sex	Coed	Single-Sex	Coed	Single-Sex	Coed	Single-Sex	Coed	Single-Sex	Coed	Single-Sex
Low	0.480	0.449	0.429	0.446	0.312	0.270	0.245	0.253	0.091	0.092	0.106	0.105
Medium	(0.500) 0.417	(0.498) 0.435	(0.495) 0.416	(0.497) 0.415	(0.463) 0.610	(0.444) 0.646	(0.430) 0.647	(0.435) 0.651	(0.288) 0.540	(0.289) 0.495	(0.308) 0.497	(0.307) 0.536
	(0.493)	(0.496)	(0.493)	(0.493)	(0.488)	(0.478)	(0.478)	(0.477)	(0.499)	(0.500)	(0.500)	(0.499)
High	0.103	0.115	0.154	0.139	0.079	0.084	0.108	0.096	0.368	0.413	0.397	0.359
Observations	(0.304) 1099.000	(0.319) 1243.000	(0.361) 1069.000	(0.346) 1345.000	(0.269) 1093.000	(0.277) 1246.000	(0.311) 1071.000	(0.295) 1342.000	(0.483) 1096.000	(0.493) 1245.000	(0.489) 1071.000	(0.480) 1345.000
		Con	fidence in N	Aath				Educat	ional Aspira	tion		
		Female		Male					Female		Male	
	Coed	Single-6	Sex (Joed Si	ngle-Sex			Coed	Single-	Sex C	loed S	ingle-Sex
Low	0.060	0.051		.039	0.041	Secondary	Education	0.040	0.035	3 0.	.045	0.046
:	(0.238)	(0.221) (((.193)	(0.197)			(0.197)	(0.178)	8) (0)	(208)	(0.210)
Medium	(0.358)	0.871	1 () ()	1.833	(0.362)	Vocational	Education	(0.254)	90.0 (0.237	9 (0 (0	.078 269)	(0.292)
High	0.090	0.078			(0.114)	Tertiary E	ducation	0.795	0.802	2 0.	769	0.747
)	(0.287)	(0.268)	3) (C	(.335)	(0.318)	•		(0.404)	(0.398)	3) (0.	(422)	(0.435)
Observations	1096.000	1246.0	00 10	71.000 1	341.000			1095.000	1247.0	00 106	7.000	1349.000

Table A.5Descriptives: Attitude towards Math

Ifo Working Papers

- No. 145 Nagl, W., Wage Compensations Due to Risk Aversion and Skewness Affection German Evidence, October 2012.
- No. 144 Triebs, T.P. and S.C. Kumbhakar, Productivity with General Indices of Management and Technical Change, October 2012.
- No. 143 Ketterer, J.C., The Impact of Wind Power Generation on the Electricity Price in Germany, October 2012.
- No. 142 Triebs, T.P., D.S. Saal, P. Arocena and S.C. Kumbhakar, Estimating Economies of Scale and Scope with Flexible Technology, October 2012.
- No. 141 Potrafke, N. und M. Reischmann, Fiscal Equalization Schemes and Fiscal Sustainability, September 2012.
- No. 140 Fidrmuc, J. and C. Hainz, The Effect of Banking Regulation on Cross-Border Lending, September 2012.
- No. 139 Sala, D. and E. Yalcin, Export Experience of Managers and the Internationalization of Firms, September 2012.
- No. 138 Seiler, C., The Data Sets of the LMU-ifo Economics & Business Data Center A Guide for Researchers, September 2012.
- No. 137 Crayen, D., C. Hainz and C. Ströh de Martínez, Remittances, Banking Status and the Usage of Insurance Schemes, September 2012.
- No. 136 Crivelli, P. and J. Gröschl, The Impact of Sanitary and Phytosanitary Measures on Market Entry and Trade Flows, August 2012.
- No. 135 Slavtchev, V. and S. Wiederhold, Technological Intensity of Government Demand and Innovation, August 2012.
- No. 134 Felbermayr, G.J., M. Larch and W. Lechthaler, The Shimer-Puzzle of International Trade: A Quantitative Analysis, August 2012.

- No. 133 Beltz, P., S. Link and A. Ostermaier, Incentives for Students: Evidence from Two Natural Experiments, August 2012.
- No. 132 Felbermayr, G.J. and I. Reczkowski, International Student Mobility and High-Skilled Migration: The Evidence, July 2012.
- No. 131 Sinn, H.-W., Die Europäische Fiskalunion Gedanken zur Entwicklung der Eurozone, Juli 2012.
- No. 130 Felbermayr, G.J., A. Hauptmann and H.-J. Schmerer, International Trade and Collective Bargaining Outcomes. Evidence from German Employer-Employee Data, March 2012.
- No. 129 Triebs, T.P. and S.C. Kumbhakar, Management Practice in Production, March 2012.
- No. 128 Arent, S., Expectations and Saving Behavior: An Empirical Analysis, March, 2012.
- No. 127 Hornung, E., Railroads and Micro-regional Growth in Prussia, March, 2012.
- No. 126 Seiler, C., On the Robustness of the Balance Statistics with respect to Nonresponse, March 2012.
- No. 125 Arent, S., A. Eck, M: Kloss and O. Krohmer, Income Risk, Saving and Taxation: Will Precautionary Saving Survive?, February 2012.
- No. 124 Kluge, J. and R. Lehmann, Marshall or Jacobs? Answers to an Unsuitable Question from an Interaction Model, February 2012.
- No. 123 Strobel, T., ICT Intermediates, Growth and Productivity Spillovers: Evidence from Comparison of Growth Effects in German and US Manufacturing Sectors, February 2012.
- No. 122 Lehwald, S., Has the Euro Changed Business Cycle Synchronization? Evidence from the Core and the Periphery, January 2012.
- No. 121 Piopiunik, M. and M. Schlotter, Identifying the Incidence of "Grading on a Curve": A Within-Student Across-Subject Approach, January 2012.
- No. 120 Kauppinen, I. and P. Poutvaara, Preferences for Redistribution among Emigrants from a Welfare State, January 2012.